Author’s response to reviews

Title: Eosinophils in anti-neutrophil cytoplasmic antibody associated vasculitis

Authors:

Thomas Hellmark (Thomas.Hellmark@med.lu.se)
Sophie Ohlsson (sophie.ohlsson@med.lu.se)
Åsa Pettersson (Asa.Pettersson@med.lu.se)
Markus Hansson (Markus.Hansson@med.lu.se)
Åsa Johansson (Asa.Johansson@med.lu.se)

Version: 4 Date: 07 Feb 2019

Author’s response to reviews:

Dear sirs,

We have now rephrased all of the indicated major overlaps as well as some other minor parts that we have identified and hope that the manuscript now is acceptable. Since there is overlap in both the cohort and methods with Johansson et al Arthritis Research and Therapy 2016: 18:92 there will always be minor overlap especially in the method section that are hard to rephrase.

Yours sincerely,

Thomas Hellmark